Author's response to reviews

Title: Memory impairment caused by cerebral hematoma in the left medial temporal lobe due to ruptured posterior cerebral artery aneurysm

Authors:

Shinichiro Maeshima (maeshima@saitama-med.ac.jp)
Aiko Osawa (aiko_o_med@yahoo.co.jp)
Fumitaka Yamane (fyamane@saitama-med.ac.jp)
Hidetoshi Shimaguchi (gucchi42@saitama-med.ac.jp)
Ikuo Ochiai (ikochi@saitama-med.ac.jp)
Ryuzaburo Kanazawa (jpn2412@saitama-med.ac.jp)
Shoichiro Ishihara (shopar72@saitama-med.ac.jp)

Version: 4
Date: 11 February 2014

Author's response to reviews: see over
Dear editor,

Thank you very much for giving us the opportunity to revise our paper. We wish to express our gratitude for your constructive comments and suggestions regarding our report. We have revised some parts of our manuscript as indicated below on the basis of your suggestion. The revised portions have been underlined in the manuscript. We greatly appreciate both your help and that of the referees concerning for improvements to this paper. We hope that the revised manuscript is now suitable for publication.

Sincerely,

Shinichiro Maeshima, M.D., Ph.D.

Reviewer 2
1. You may want to include for reference some previous reports regarding mild or no retrograde amnesia after lesion restricted to the left medial temporal lobe.
   →We have cited previous reports which you instructed in the discussion and added references in the references section (underlined).

2. The order of the results of neuropsychological test in Table 1 can be revised with listing three types of memory tests together, not separately. If possible, adding cutoff point or mean plus/minus deviation for each test would be informative for readers.
   →As you instructed, we have revised Table 1 with listing three types of memory test together, and added mean (SD) or cut-off points in some neuropsychological tests.